CMSC 828G Project Part 2

Data Mining,CMSC 828G, Spring 2002

Due Date Thursday March 14 in class PLEASE BRING THREE COPIES OF YOUR PROPOSAL Tuesday 19 in class PLEASE BRING TWO COPIES OF EACH OF YOUR REVIEWS

Project Proposal

Please write a two page description of your project. Please turn in three copies. You should:

• Briefly describe the problem domain.

• Describe your data mining goals.

• Describe the hypothesis being tested and the experimental setup.

• Describe the algorithms you will use. If you intend to use existing software, describe the software. If you plan to write your own software, please scope out the task. Please be specific.

• Describe your evaluation criteria.

• Provide milestones for each week from now until the project is due (4/25).

Peer Review

In class on Thursday 3/14, each person will receive copies of two of the other student’s proposals. Your job is to write a 1/2 page to 1 page review of each proposal. Your goal in reviewing is to make suggestions that will improve the quality of the project. All feedback should be positive and constructive. Your job is to help the other student’s project to be well-defined, relevant and interesting.

On Tue. 3/19, You will turn in two copies of each of your two reviews. You will be graded on the quality of your suggestions.